Author’s response to reviews

Title: Red wine consumption increases antioxidant status and decreases oxidative stress in the circulation of both young and old humans

Authors:

Michelle A Micallef (michelle.micallef@studentmail.newcastle.edu.au)
Louise A Lexis (louise.lexis@vu.edu.au)
Paul A Lewandowski (paul.lewandowski@deakin.edu.au)

Version: 6 Date: 4 September 2007

Author’s response to reviews:

To Whom It May Concern:

The following minor alterations were made to the manuscript ¿Red wine consumption increases...¿ (Manuscript ID 1533966001126051).

Table 1 had a correction in the formatting of the value for total phenolics, there was an additional space in front of the 40 a.u. value which effected its alignment in the table column that it was contained in.

Figures 1 and 2 have been cropped further to reduce the amount of white space around each figure.

Regards,

Paul